Author's response to reviews

Title: Cytotoxicity of VEGF121/rGel on Vascular Endothelial Cells Resulting in Inhibition of Angiogenesis is Mediated via VEGFR-2

Authors:

Khalid A Mohamedali (kmohamed@mdanderson.org)
Sophia Ran (sran@siumed.edu)
Candelaria Gomez-Manzano (cmanzano@mdanderson.org)
Latha Ramdas (LRamdas@mdanderson.org)
Jing Xu (Jing.Xu@UTSouthwestern.edu)
Sehoon Kim (SehoonKim@sbcglobal.net)
Lawrence H Cheung (LCheung@mdanderson.org)
Walter N Hittelman (Whittelman@mdanderson.org)
Wei Zhang (WZhang@mdanderson.org)
Johannes Waltenberger (waltenberger@ukmuenster.de)
Philip E Thorpe (Philip.Thorpe@UTSouthwestern.edu)
Michael G Rosenblum (mrosenbl@mdanderson.org)

Version: 6 Date: 12 August 2011

Author's response to reviews: see over
August 12, 2011

The Editors
BMC Cancer
BioMed Central Ltd,
Floor 6, 236 Gray’s Inn Road
London, WC1X 8HB
United Kingdom

Re: MS 1605395934542378
Cytotoxicity of VEGF_{121/7Gel on Vascular Endothelial Cells Resulting in Inhibition of Angiogenesis is Mediated via VEGFR-2 by Khalid A. Mohamedali et al.

Dear Editors,

As part of the final preparation of our manuscript for publication, all formatting changes have been made and the manuscript has been proof-read for errors as requested by the BMC Series Editorial Team. Specifically:

- Countries have been included as part of the author affiliation.
- The references have been modified to not include the term ‘et al.’ unless there are 30 or more authors.
- Highlighting/tracking has been removed from the system.
- All vertical lines in Table 1 have been removed.
- All figures have been closely cropped.

On behalf of my colleagues, thank you very much for your acceptance of our manuscript in BMC Cancer.

Sincerely,

Khalid A. Mohamedali, Ph. D.
Department of Experimental Therapeutics
University of Texas M.D. Anderson Cancer Center
1515 Holcombe Blvd, Unit 1950
Houston, TX 77030
Tel: 713-792-5954
kmohamed@mdanderson.org